Author's response to reviews

Title: Developmental Trajectories of Physical Activity and Television Viewing during Adolescence among Girls: National Growth and Health Cohort Study

Authors:

Soyang Kwon (skwon@luriechildrens.org)
Jungwha Lee (jungwhalee@northwestern.edu)
Mercedes Carnethon (mcarnethon@northwestern.edu)

Version: 4 Date: 1 July 2015

Author's response to reviews:

Dear BMC Public Health Editors,

To address the request of including the full names and affiliations of all the ethic boards that approved this study, in this revision, the authors have added the following statements in Methods: “The NGHS was approved by University of California-Berkeley Institutional Review Boards (IRB), University of Cincinnati IRB, and Westat IRB. The current study of de-identified existing data was exempt from Ann & Robert H. Lurie Children’s Hospital of Chicago IRB.”

If you have any additional questions or concerns, please do not hesitate to contact me.

Sincerely,

Soyang Kwon, PhD
Research Assistant Professor, Stanley Manne Children’s Research Institute
Ann & Robert H. Lurie Children’s Hospital of Chicago
Northwestern University Feinberg School of Medicine
T 312.227.7033 | F 312.227.9523 | skwon@luriechildrens.org | luriechildrens.org
225 East Chicago Avenue, Box 157, Chicago, Illinois 60611-2991